Author's response to reviews

Title: Derivation and Validation of an Accurate Estimation of CD4 counts from the Absolute Lymphocyte Count in Virologically Suppressed and Immunologically Reconstituted HIV Infected Adults

Authors:

Barnaby E Young (barnaby.young@gmail.com)
David C Lye (david_lye@ttsh.com.sg)
Oon Tek Ng (oon_tek_ng@ttsh.com.sg)
Yee Sin Leo (yee_sin_leo@ttsh.com.sg)

Version: 4 Date: 31 July 2015

Author's response to reviews:

1. Database access

Please clarify in the manuscript whether access to the databases you used is freely available, and if not, who gave permission to use them. Please list the permissions for each database.

The following has been added to the final line of the methods section:

'... and access to the database approved by the Head of Infectious Diseases Department, Tan Tock Seng Hospital.'